Author’s response to reviews

Title: The Need for Education on Health Related-Quality of Life

Authors:

Melanie J Calvert (m.calvert@bham.ac.uk)
John R Skelton (J.Skelton@bham.ac.uk)

Version: 2 Date: 19 November 2007

Author’s response to reviews: see over
Dear Editor,

We hope you will be interested in publishing the enclosed article: “The Need for Education on Health Related-Quality of Life” in BMC Medical Education. Health-related quality of life is increasingly recognised as an important outcome measure, and the need for education on this subject has been identified. Education available on this subject for different healthcare professionals is varied. This paper describes the development and implementation of a quality of life special study module which provided undergraduate medical students with a new opportunity to develop their understanding of quality of life research and practice. The paper also highlights an education initiative currently under development by the International Society for Quality of Life.

Yours Sincerely,

Melanie Calvert, PhD

Research Fellow
Department of Primary Care and General Practice
Tel: +44 (0) 121 414 8595
Fax: +44 (0)121414 6571
Email: M.Calvert@bham.ac.uk